

Author's response to reviews

Title: An investigation of the use of acupuncture in stroke patients in Taiwan: a national cohort study

Authors:

Shu-Wen Weng (aaaa6712@gmail.com)

Ta-Liang Chen (tlc@tmu.edu.tw)

Chun-Chieh Yeh (b8202034@gmail.com)

Chien-Chang Liao (jacky48863027@yahoo.com.tw)

Hsin-Long Lane (imcold0404@hotmail.com)

Jaung-Geng Lin (jglin@mail.cmu.edu.tw)

Chun-Chuan Shih (hwathai@seed.net.tw)

Version: 2 Date: 26 Jul 2016

Author's response to reviews:

Response to editors:

1

Q: Thank you for responding to our request to include a Declarations section in your manuscript. However, we would ask you to please move this to after your Conclusions section, in accordance with our formatting guidelines. We would also ask you to please include each of the required sub-sections in this section, as detailed in our formatting guidelines (<http://bmccomplementalrmed.biomedcentral.com/submission-guidelines/research-article>). In the event that any of these sections are not relevant for your study, we would still ask you to include them, but to include a statement detailing that this information is not relevant.

Reply: By your comment, the Declarations section was removed to after Conclusion section and all of them with subheading. Thanks.

2.

Q: Please remove the tracked-changes from your manuscript. While we understand that these have been included in order for you to indicate where changes have been made in your manuscript, and we appreciate this, they have been known to cause issues with our Production systems.

Reply: By your comment, the tracked-changes was removed. Thanks.